Author's response to reviews

Title: Solid variant of aneurysmal bone cyst of the heel: a case report

Authors:

Joanna A Lekka (Jlekka@yahoo.gr)
Theofani V Gavressea (fangav@261.gr)
Gabriella A Stanc-Giannakopoulos (gabidr1999@yahoo.com)
Nikolaos S Demertzis (NDemertzis@yahoo.gr)

Version: 11 Date: 12 January 2011

Author's response to reviews: see over
Author's response to reviews

Solid variant of aneurysmal bone cyst of the heel: a case report

Joanna A. Lekka (not@valid.com)
Th. Gavresea (fangav@yahoo.gr)
G.Stanc-Giannakopoulos (gabidr1999@yahoo.com)
N. Demertzis (not@valid.com)

Author response to review: see over

Restructured abstract:
Editor's report

Major revisions (we require the author to make these changes)

1. Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

Changes have been made as you indicate.

2. Authors' contributions: Please indicate that all authors read and approved the final manuscript. Please ensure this is formatted into a paragraph rather than a list.

Changes have been made as you indicate.

3. Structure: Please reformat the end of the manuscript so that it follows the following structure: Conclusions, Abbreviations (if used), Competing interests, Consent, Authors' contributions, Acknowledgements (optional), References

Changes have been made as you indicate.

4. A change has been made in the e-mails of the authors

Version 10: 12 January 2011